Author's response to reviews

**Title:** Alcohol assessment & feedback by e-mail for university student hazardous and harmful drinkers: study protocol for the AMADEUS-2 randomised controlled trial

**Authors:**

Jim McCambridge ([Jim.McCambridge@lshtm.ac.uk](mailto:Jim.McCambridge@lshtm.ac.uk))
Marcus Bendtsen ([marcus.bendtsen@liu.se](mailto:marcus.bendtsen@liu.se))
Nadine Karlsson ([nadine.karlsson@liu.se](mailto:nadine.karlsson@liu.se))
Ian R White ([ian.white@mrc-bsu.cam.ac.uk](mailto:ian.white@mrc-bsu.cam.ac.uk))
Preben Bendtsen ([preben.bendtsen@liu.se](mailto:preben.bendtsen@liu.se))

**Version:** 2 **Date:** 21 August 2013

**Author's response to reviews:** see over
Cover letter for submission of ‘Alcohol assessment & feedback by e-mail for university student hazardous and harmful drinkers: study protocol for the AMADEUS-2 randomised controlled trial’ to BMC Public Health

This trial protocol is for a study in an area of clear importance to public health. We anticipate that there will be wide interest in the detail of the design of this study. No other articles relating to the protocol have been published or submitted elsewhere for consideration. We have made the requested changes to MS: 9098120301039085 in the e-mail dated 29/07/13.

Jim McCambridge